


9th ATF2 Project Meeting

Thursday 17 December 2009

Updates of commissioning status - Seminar Hall, 3-Go-Kan (13:00 - 15:20)

time [id] title presenter

13:00 [100] Optics Modeling and BPM status Mr RENIER, Yves

13:20 [101] IP tuning Ms BAI, Sha

13:40 [99] Laserwire OTR Study Report Dr KARATAEV, Pavel

14:00 [104] IP BPM Electronics Beam Test Ms KIM, Youngim

14:20 [102] Cavity BPM progress in this week Dr BOOGERT, Stewart

14:40 [103] Beam tuning progress in this week Dr WHITE, Glen
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